Author's response to reviews

Title: Anti-inflammatory effect of ethanolic extract from Myagropsis myagroides on murine macrophages and mouse ear edema

Authors:

Eun-Ji Joung (eunji2007@naver.com)
Min-Sup Lee (minwill@hanmail.net)
Ji-Woong Choi (antichem@hanmail.net)
Jong-Soon Kim (rosajsysang@yahoo.co.kr)
Taisun Shin (shints@chonnam.ac.kr)
Bok-Mi Jeong (jbm@chonnam.ac.kr)
Na Young Yoon (dbssud@nfrdi.go.kr)
Chi-Won Lim (cwlim@nfrdi.go.kr)
Jae-II Kim (jikim@pknu.ac.kr)
Hyeung-Rak Kim (hkim@pknu.ac.kr)

Version: 2 Date: 23 May 2012

Author's response to reviews: see over
Re: MS # 7585660077348853

Dear Editor:

Thank you very much for your e-mail of May 23, 2012, with regard to our manuscript entitled “Anti-inflammatory effect of ethanolic extract from Myagropsis myagroides on murine macrophages and mouse ear edema” (MS # 7585660077348853) together with “Editorial requests”.

As suggested, we have included more detailed information on the ethical approval of our animal protocols and procedures into the Methods section (lines 3-6 in the third paragraph on page 11).

I would be grateful if the manuscript could be reviewed and considered for publication in BMC Complementary & Alternative Medicine as a Research article. This manuscript contains original work and that it has not been published or submitted for publication elsewhere.

We look forward to hear from you on the disposition of the manuscript. Thank you for your consideration.

Sincerely,

Hyeung-Rak Kim, Ph.D
Department of Food Science and Nutrition
Pukyong National University
Busan 608-737, South Korea
Phone: 82-51-629-5847
Fax: 82-51-629-5842
E-mail: hrkim@pknu.ac.kr